Author's response to reviews

Title: The Impact of Direct Provision Accommodation for Asylum Seekers on Organisation and Delivery of Local Primary Care and Social Care Services: A Case Study

Authors:

Hans-Olaf Pieper (hopieper@hotmail.com)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Anne MacFarlane (anne.macfarlane@nuigalway.ie)

Version: 6 Date: 4 May 2011

Author's response to reviews:

May 4th 2011

Dear colleagues,

Thank you for your e-mail about the final revisions for MS 3116265374630609.

The following requested changes have been made:

I have provided functioning links to website reference numbers 7, 11, 18.

I have included full details of reference 41 are now included as this article was published this week.

On the title page, 'Title:' has been removed.

The visible vertical lines have been removed from the tables.

The figures have been cropped as closely as possible to minimise white space around the images.

The manuscript has been checked for any typographical errors.

Thank you for your support with this publication.

Best wishes,
Anne MacFarlane